Author's response to reviews

Title: Communicating serum chemical concentrations to study participants: follow up survey

Authors:

Alexandra J Buck (ajbuck@buffalo.edu)
John E Vena (jvena@uga.edu)
Bridget M McGuinness (bridget.m.mcguinness@gmail.com)
Maureen A Cooney (cooneyma@mail.nih.gov)
Germaine M Louis (louisg@mail.nih.gov)

Version: 4 Date: 21 April 2010

Author's response to reviews: see over
Dear Editor:

Enclosed please find our revised paper entitled *Communicating Individual Chemical Exposures to Study Participants* that we are submitting for final consideration by *Environmental Health*. We have made all requested edits as the final stage of proofing.

As requested, a point-by-point response follows.

Respectfully submitted,

Maureen A. Cooney, PhD, MPH
Corresponding Author
Formatting changes:
1. The Abstract is present in full in the manuscript and matches exactly what is in the submission system.
2. A sentence to the authors’ contributions has been added to indicate all authors read and approved the final manuscript
3. The references have been edited and are now in sequence

Minor revisions

1. Countries are included in the affiliations of authors
2. Highlighting has been removed
3. Abbreviations have been formatted as a paragraph